Author's response to reviews

Title: Twenty-four hours secretion pattern of serum estradiol in healthy prepubertal and pubertal boys as determined by a validated ultra-sensitive extraction RIA

Authors:

Carina Ankarberg-Lindgren (carina.ankarberg-lindgren@vgregion.se)
Ensio Norjavaara (ensio.norjavaara@vgregion.se)

Version: 11 Date: 15 September 2008

Author's response to reviews:

We have made the following corrections:
1) inserted a conclusion and abbreviation section.
2) removed all unnecessary capitals.
3) revised the reference list.